Author’s response to reviews

Title: Self-efficacy and self-rated oral health among pregnant Aboriginal Australian women

Authors:

Lisa M. Jamieson (lisa.jamieson@adelaide.edu.au)
Eleanor J. Parker (eleanor.parker@adelaide.edu.au)
Kaye F. Roberts-Thomson (kayeroberts-thomson@adelaide.edu.au)
John S. Broughton (john.broughton@otago.ac.nz)
Herenia P. Lawrence (herenia.lawrence@dentistry.utoronto.ca)

Version: 4 Date: 17 March 2014

Author’s response to reviews:

2nd revisions:

1. Extra ‘pregnant’ removed from section in Abstract Background.
2. ‘Written informed consent was received’ added to Methods Page 5, Paragraph 2
3. We are not allowed to upload the questionnaire due to the University of Adelaide’s Intellectual Property policies.